Author’s response to reviews

Title: How residents and interns utilise and perceive the personal digital assistant and UpToDate

Authors:

Jason Phua (phua_jason@yahoo.com.sg)
Tow Keang Lim (mdclimtk@nus.edu.sg)

Version: 5 Date: 3 July 2008

Author’s response to reviews: see over
3 Jul 2008

To: Dr Hans Zauner, PhD
Assistant editor
BMC-series journals
BMC Medical Education

Dear Dr Zauner,

**Manuscript: How residents and interns utilise and perceive the personal digital assistant and UpToDate**

Thank you for accepting our manuscript.

As requested, we have made the necessary formatting changes in the following areas: author list, affiliations, abbreviations, figure titles, figure cropping and additional files. To simplify matters, we have opted to remove all Endnote field codes and are prepared to forego the discount offered for the use of Endnote.

Yours sincerely,

Dr Phua Jason
Professor T. K. Lim
Department of Medicine
National University Hospital
Singapore